Author’s response to reviews

Title: Intravitreal dexamethasone implant for central retinal vein occlusion without macular edema

Authors:

Eun Choi (eunyoung.choi86@gmail.com)

Hyun Kang (HGKANG08@yuhs.ac)

Sung Lee (SUNGLEE@yuhs.ac)

Min Kim (minkim76@gmail.com)

Version: 1 Date: 24 Jan 2019

Author’s response to reviews:

We have revised the present paper according to your useful suggestions and comments. Thank you.

1. We note that the manuscript and its figure legends contain potentially identifiable information. Currently, the exact age and gender information, in combination with other identifiers, may compromise patient/participant anonymity.

   Please amend the manuscript and figure legends to address this issue, by removing this information and providing ages as age-ranges (20-29, 30-39 etc).

   Response> We have modified the exact age to the age range to remove potentially identifiable information.

2. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

   Response> We have added “Figure Legends” section under a separate heading.

3. Please add a section entitled "Keywords" after the conclusion section of the Abstract. This should contain three to ten keywords representing the main content of the article.

   Response> We have moved "Keywords" section after the conclusion section of the Abstract.